Author’s response to reviews

Title: Secondary breast lymphoma diagnosed by Vacuum-assisted breast biopsy: case report.

Authors:

Flora Zagouri (florazagouri@yahoo.co.uk)
Theodoros N. Sergentanis (tsergentanis@yahoo.gr)
Afroditi Nonni (afnonni@med.uoa.gr)
Dimitra Koulocheri (dim.koul@altecnet.gr)
Philip Domeyer (philip.domeyer@gmail.com)
Dimitrios Dardamanis (dardaman@hol.gr)
Nikolaos V. Michalopoulos (nmichal@med.uoa.gr)
Nikolaos Pararas (npararas@med.uoa.gr)
Antonia Gounaris (antgouna@otenet.gr)
George C. Zografos (gzografo@med.uoa.gr)

Version: 3 Date: 28 August 2007

Author's response to reviews: see over
George C. Zografos  
Assoc. Professor of Surgery, University of Athens  
101, Vas Sofias Ave,  
Ampelokipi, Athens 11521, Greece.  
e-mail: gzografo@med.uoa.gr  
tel. number: +302106426390  
fax: + 302106426390  
Mobile: +306932524836

To  
The Editorial Board of The Journal of Medical Case Reports  

Athens, 28/8/2007

Dear Editor,  

I am sending for publication to your distinguished Journal the revised version of the case report entitled: “Secondary breast lymphoma diagnosed by Vacuum-assisted breast biopsy: case report.”  

Within the next page, a list of changes is provided, according to the reviewers’ comments.  

Yours sincerely,  

George C. Zografos,  
Assoc. Professor of Surgery  
University of Athens
**List of changes**

**Reviewer 1 (Michael Stark)**
The manuscript has been checked again for spelling, and all errors have been corrected. Where needed, English has been revised.

**Reviewer 2 (Jean Karl Soler)**
- The decimal points have been deleted in the patient’s BMI (case presentation, paragraph 3, line 4).
- BI-RADS 4B is explained in the revised version of the manuscript according to the standard definition (case presentation, paragraph 4, line 2).